Author’s response to reviews

Title: A Conceptual Model of Barriers and Facilitators to Primary Clinical Teams Requesting Pediatric Palliative Care Consultation Based Upon a Narrative Review

Authors:

Jennifer Walter (walterj1@email.chop.edu)

Douglas Hill (hilld@email.chop.edu)

Concetta DiDomenico (didomenico@email.chop.edu)

Shefali Parikh (parikhsh@email.chop.edu)

Chris Feudtner (feudtner@email.chop.edu)

Version: 2 Date: 26 Nov 2019

Author’s response to reviews:

Thank you to the reviewers for the additional comments and opportunity to respond. The reviewer's comments are below followed by the response.

1) in the methods section on page 3, it would be helpful if the authors could add references to the 'social psychology theories of individual and group behavior'

We have added the appropriate references in the Methods section on p3 lines 13 and updated the references in the Figure 1 to correspond.

2) in the abstract, facilitator 5 is still described as 'shared serious news within the team' which should be 'sharing serious news as a team'

We have made the edit in the abstract on p2 line 24.